Author’s response to reviews

Title: Hemophagocytic lymphohistiocytosis resulting from a cytokine storm triggered by septicemia in a child with chronic granuloma disease: A case report and literature review

Authors:

Ang Wei (syweiang@126.com)
Honghao Ma (723635372@qq.com)
Liping Zhang (zlp2015zlp@163.com)
Zhigang Li (ericlzg70@hotmail.com)
Qing Zhang (zhangqing19900210@126.com)
Dong Wang (619608570@qq.com)
Li Zhang (syweiang@163.com)
Hongyun Lian (hongyun_lian73@aliyun.com)
Rui Zhang (ruizh1973@126.com)
Tianyou Wang (wangtianyou@bch.com.cn)

Version: 1 Date: 19 Jan 2020

Author’s response to reviews:

To the editor:

Thanks for the reviewer’s comments and advice, I have made some changes according to the reviewer’s advice. The modifications are listed below:

1. This manuscript has been rewritten by a native English speaker.
2. The main message of this case report has been simplified. (Revise Page 1, Line 21-24. Delete Page 4, Line 2-3; Page 4, Line 5-16; Page 5, Line 3, 4-11, 13-14, 16-19, 28. Page 6, Line 1-2, 4-5)
3. Cytokine screening can help physician to distinguish sepsis from HLH. At the same time, cytokines can also be used as important indicators to monitor the activation of HLH. And they can be detected by flow cytometry.
4. The literature and following issues recommended by the Reviewer 2 has been added in the section of references. And the order of references has also been adjusted. (Page 2, Line 15; Page 5, Line 15)

With Kind Regards
Ang Wei